Dear Editor,

Thank you for considering the manuscript entitled “Gastric Wall Necrosis: A Rare Complication of Endoscopic Treatment for Bleeding Gastric Ulcer” (91370). We really appreciate all the valuable comments and constructive suggestions from reviewers. We have revised the manuscript and a point-by-point response was enclosed. We would like to re-submit the revised manuscript to the World Journal of Gastrointestinal Surgery, and hope it is acceptable for publication in the journal. Please do not hesitate to contact us for any question or concern.

We look forward to your final decision.

Sincerely yours

Xi-qiu Yu
Responses to reviewer:

1. The authors report a case of a bleeding giant gastric ulcer complicated by subsequent gastric wall necrosis. Injection of sclerosants and tissue glue may lead to gastric wall necrosis, which is a serious complication. Therefore, before administering this treatment to patients, we should consider other more effective methods of hemostasis to avoid gastric wall necrosis. It would be interesting to include a short review of the literature on this problem.

Response: Thank you. We have supplemented the points and revised the manuscript.